Author’s response to reviews

Title: Hospital ethics reflection groups: a learning and development resource for clinical practice

Authors:

Henriette Bruun (henriette.bruun@rsyd.dk)

Lotte Huniche (LHuniche@health.sdu.dk)

Elsebehlt Stenager (Elsebeth.Stenager@rsyd.dk)

Reidar Pedersen (reidar.pedersen@medisin.uio.no)

Version: 4 Date: 06 Oct 2019

Author’s response to reviews:

Dear Janelle Coore

This is a point-by-point response letter to the revision required

1. Please confirm whether the consent was written or verbal in the Ethics approval and consent to participate and Methods sections. If verbal, please confirm that your ethics committee formally approved this consent and how the verbal consent was documented.

I´m sorry for overlooking your request of confirming, that the consent of participation was written. This correction has been made in both the Ethics approval and consent to participate and Method sections

Best wishes

Henriette Bruun